**Author’s response to reviews**

**Title:** Barriers to the acceptance of electronic medical records by physicians  
From systematic review to taxonomy and interventions

**Authors:**

Albert Boonstra (albert.boonstra@rug.nl)  
Manda Broekhuis (h.broekhuis@rug.nl)

**Version:** 5  **Date:** 6 August 2010

**Author’s response to reviews:**

Dear members of the production team,

The title of the paper is:

Barriers to the acceptance of electronic medical records by physicians.

The subtitle of the paper is:

From systematic review to taxonomy and interventions.